Author's response to reviews

Title: Assessing the disease burden of Yi people by Years of Life Lost in Shilin county of Yunnan province, China

Authors:

Shang-Cheng Zhou (zsc_2k@163.com)
Le Cai (caile002@hotmail.com)
Chong-Hua Wan (wanchh@hotmail.com)
Yi-Ling Lv (zjc@yymc.edu.cn)
Peng-Qian Fang (pfang@mails.tjmu.edu.cn)

Version: 9 Date: 18 May 2009

Author's response to reviews:

To editors of BMC Public Health
Dear Sir or Madam:
I am submitting the revised manuscript “Assessing the disease burden of Yi people by years of life lost in Shilin county of Yunnan province, China” (MS: 1973280393251826) to BMC Public Health.
I have checked over your manuscript carefully and revised some typographical errors.
Thank you for the kind advice.
Best wishes.
Zhou Shang Cheng